

# GOPEN ACCESS

**Citation:** Shirley D-A, Moonah S (2016) Fulminant Amebic Colitis after Corticosteroid Therapy: A Systematic Review. PLoS Negl Trop Dis 10(7): e0004879. doi:10.1371/journal.pntd.0004879

Editor: Isaura Meza, CINVESTAV, MEXICO

Received: May 17, 2016

Accepted: July 6, 2016

Published: July 28, 2016

**Copyright:** © 2016 Shirley, Moonah. This is an open access article distributed under the terms of the <u>Creative Commons Attribution License</u>, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the paper and its Supporting Information file.

**Funding:** SM received funding from NIH grants R01Al026649 and K08Al119181. DAS received no specific funding for this work. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

**Competing Interests:** The authors have declared that no competing interests exist.

**RESEARCH ARTICLE** 

# Fulminant Amebic Colitis after Corticosteroid Therapy: A Systematic Review

#### Debbie-Ann Shirley<sup>1</sup>, Shannon Moonah<sup>2</sup>\*

1 Department of Pediatrics, University of Virginia School of Medicine, Charlottesville, Virginia, United States of America, 2 Department of Medicine, University of Virginia School of Medicine, Charlottesville, Virginia, United States of America

\* sm5fe@virginia.edu

# Abstract

## Background

Amebic colitis, caused by intestinal infection with the parasite, *Entamoeba histolytica*, is a common cause of diarrhea worldwide. Fulminant amebic colitis is the most devastating complication of this infection, associated with both high mortality and morbidity. We conducted a review of the English literature to describe cases of fulminant amebic colitis associated with exposure to corticosteroid medications in order to identify the risk factors for poor outcome and determine difficulties in diagnosis and treatment.

# **Methodology and Principal Findings**

Articles reporting severe and fulminant forms of amebic colitis between 1991 and 2016 were collected. 525 records were screened to identify 24 cases for qualitative analysis associated with corticosteroid use. Cases arose from areas of high endemicity or travel to such areas. Most cases (14 of 24, 58%) were given corticosteroids for initially misdiagnosed colitis, mainly inflammatory bowel, resulting in rapid progression of disease. Nearly half of all cases underwent surgical intervention, and 25% of cases died, despite all patients eventually receiving treatment with metronidazole. The odds of death did not differ significantly by prior misdiagnosis, co-morbidities, bowel perforation or need for surgery.

# **Conclusions and Significance**

Infection with *E*. *histolytica* should be considered prior to the administration of corticosteroids, in particular for patients residing in endemic areas or those with appropriate travel history, especially prior to the diagnosis of inflammatory bowel disease. The development of preventative and treatment interventions are needed to improve outcomes of fulminant disease.

# Author Summary

*Entamoeba histolytica* is one of the most important enteric pathogens affecting people worldwide, causing the diarrheal disease amebic colitis. Fulminant amebic colitis is an uncommon but life threatening complication that may ensue. High rates of colonization

with *E. histolytica* burden many developing countries, and travelers are at risk of acquisition of infection when they visit endemic areas. Corticosteroids are an invaluable group of broadly prescribed anti-inflammatory medications, but have been identified as a risk factor for the development of fulminant amebic colitis. Our comprehensive report highlights the frequent misdiagnosis of amebic colitis and the high morbidity and mortality associated with fulminant disease. Improved awareness of this condition among medical providers is needed, so that infection with *E. histolytica* can be considered in both patients presenting with symptoms of colitis and patients with asymptomatic colonization prior to the administration of corticosteroids. This study points to the need to continue efforts to develop both a vaccine that can prevent amebic colitis and innovative life- and bowel-saving adjuncts for the treatment of fulminant amebic colitis through an improved understanding of host responses to infection with *E. histolytica*.

# Introduction

Amebic colitis, caused by the protozoan parasite *Entamoeba histolytica*, is a leading cause of severe diarrhea worldwide, killing more than 55,000 people globally each year [1]. The greatest burden of amebic disease occurs in developing countries, likely due to poorer socioeconomic conditions, decreased sanitation and reduced hygiene [2]. In the largest global study of childhood diarrheal illness conducted to date, *E. histolytica* was shown to be a top cause of severe diarrhea among infants and children living in Africa and Asia, and was the leading cause of unadjusted mortality in the 12 to 24 month age group [3]. The parasite has a two-stage life cycle, existing as either an infectious cyst or invasive trophozoite. Transmission occurs via ingestion of cysts, most commonly by fecally contaminated food or water, though direct fecaloral transmission through sexual contact is also described [2,4]. Trophozoite invasion of the intestinal mucosa leading to mucosal inflammation is a hallmark of amebic colitis.

The vast majority of amebic infections are asymptomatic, with approximately 10% of those infected progressing to have symptoms. Amebic colitis is the most common symptomatic manifestation, with variable presentation, including watery diarrhea, dysentery, abdominal pain, tenderness and rarely the formation of a tumor like granulation mass referred to as an ameboma [2,5]. Trophozoites can disseminate to the liver, causing amebic abscesses, as well as to the central nervous system and other extra-intestinal sites. Fulminant amebic colitis, though uncommon, is the most serious and life-threatening complication of amebiasis, presenting initially with bloody diarrhea, fever, leukocytosis and abdominal pain. Bowel necrosis, toxic megacolon, perforation and peritonitis may ensue. Fulminant amebic colitis is associated with high mortality and morbidity, with case fatality rates ranging from 40% to 89% [6-10].

Corticosteroids are commonly prescribed in medicine for their anti-inflammatory and immunosuppressive therapeutic properties. Case reports have indicated that corticosteroids are a risk factor predisposing to the development of fulminant amebic colitis [11-13]. Given the widespread use of corticosteroids, this observation holds significant implications not only to those living in endemic areas, but with expanding patterns in travel and migration, also poses an emerging health threat to those living in more industrialized settings. Our aim was to systematically review recent articles reporting fulminant amebic colitis in patients treated with corticosteroids to identify the main risk factors for poor outcome and highlight challenges regarding diagnosis and treatment.

# Methods

We carried out a systematic review of articles published in the English literature between January 1991 and May 2016. The search was performed electronically in PubMed to find all articles

reporting amebic colitis, using the following strategy: disease ("amoebic colitis"[All Fields] OR "dysentery, amebic"[MeSH Terms] OR ("dysentery"[All Fields] AND "amebic"[All Fields]) OR "amebic dysentery"[All Fields] OR ("amebic"[All Fields] AND "colitis"[All Fields]) OR "amebic colitis"[All Fields]) AND (("199101/01"[PDAT]: "2016/05/01"[PDAT]) AND "humans"[MeSH Terms]). The search was last conducted on May 5, 2016. A similar search was performed using Google Scholar to find additional articles that may not have been cataloged in PubMed. Case reports of intestinal amebiasis were then then reviewed to determine corticosteroid exposure, including administration of systemic preparations (prednisone, prednisolone, methylprednisolone, triamcinolone, dexamethasone, cortisone acetate and hydrocortisone) as well as enema preparations (budesonide). Evidence of amebiasis included identification by stool study, tissue examination or serology. Fulminant colitis was defined as severe abdominal pain, dysentery, fever, peritonitis, perforation or the need for urgent surgical intervention [14]. When possible, odds ratios were calculated to measure association between exposure and outcome. A twotailed p-value of <0.05 was considered to be statistically significant and was determined by use of STATA, version 11 (StataCorp).

# Results

Our search strategy identified 514 publications. Eleven additional publications were identified using the alternative search engine, excluding duplicate publications. By evaluation of the title and abstract, 404 publications were excluded. Full text publications were then reviewed for eligibility including reports of intestinal amebiasis and concomitant administration of corticosteroid therapy. Among the 122 publications reviewed, a total of 23 publications, with 24 cases were found in the literature (Fig 1).

# Demographics

Cases were reported from many regions of the world, including those with both high endemicity and non-endemicity of amebiasis (Table 1). The majority of reports arose from Asia (11 of 24 cases, 46%) including Japan [15–19], Taiwan [20,21], India [22], Singapore [23], Pakistan [24] and South Korea [25]. This was followed by European countries (6 of 24 cases, 25%), including Spain [26,27], Sweden [28], Denmark [29], Portugal [30] and Italy [31]. There were three reports (12%) originating from the USA [32–34] and a fourth (4%) from the USA territory American Samoa [35]. Two cases (8%) from a single case series were from Chile [36]. There was a single case (4%) from Israel, representing the Middle East [37]. There were no reports from Africa, Central America or the Caribbean. Many cases arising from countries with low or no endemicity postulated travel to endemic countries as the most likely risk factor for transmission of infection. Reports arising from European countries reported travel to Bali [29], the Philippines [26], Brazil, Saudi Arabia, Sri Lanka, Tanzania [28], Angola [30], the Far East and North Africa [27]. Travel history of the partner was noted to be the most likely source for one case [27]. Among reports arising from the USA, travel to Mexico was listed as a significant risk factor [33]. Another report arising from the USA did not report travel history, but contaminated well water was suspected to be the source for transmission of infection [34]. Cases aged in range from 1-83 years with a median age of 41 years (Table 1), including three pediatric cases [35,36].

#### Indications for corticosteroid use

In all, 14 of 24 (58%) cases were given corticosteroid medications for initially misdiagnosed colitis (Table 1), including misdiagnosed inflammatory bowel disease (11 of 24 cases, 46%) [18–20,22,26,28–30,33,37], intestinal vasculitis (2 of 24 cases, 8%) [23,25], and abdominal tuberculosis (1 of 24 cases, 4%) [24]. The remainder were given corticosteroid medications for



Fig 1. PRISMA flow chart: Data collection and selection of case studies.

doi:10.1371/journal.pntd.0004879.g001

the indication of prevention or treatment of graft-versus-host disease (GVHD) following transplantation (5 of 24, 21%) [<u>16,31,34,36</u>], respiratory illness (3 of 24 cases, 12.5%) [<u>21,32,35</u>], multiple myeloma (1 of 24, 4%) [<u>17</u>] and thrombocytopenic purpura (1 of 24, 4%) [<u>27</u>].

Co-morbidities in this series were high (Table 1) and included 4 cases (17%) with underlying autoimmune disease [23,25,27,29]; 3 cases (12.5%) who underwent either stem cell or solid organ transplantation [16,31,34] and 2 cases (8%) with Human Immunodeficiency virus (HIV) co-infection [20,21]. Histologic evidence of cytomegalovirus (CMV) colitis was present in both of the cases with HIV, as well as a third case without additional immunocompromising state reported [20,21,25]. Eight cases (33%) had no underlying co-morbidity reported. Fourteen cases (58%) were treated with other immune modulating therapies in addition to corticosteroids. These included the anti-inflammatory agents sulfasalazine and mesalamine (8 of 24, 33%) [15,18–20,22,28,33,37], antimetabolites azathioprine or mercaptopurine (2 of 24, 8%) [34,37], methotrexate (3 of 24, 12.5%) [16,29,31], tacrolimus (2 of 22, 8%) [16,34], cyclosporine (3 of 24, 12.5%) [31,36], anti-thymocyte antiglobulin (1 of 24, 4%) [16] and hydroxychloroquine (1 of 24, 4%) [25]. The case with thrombocytopenic purpura underwent plasmapheresis while also receiving corticosteroids and prior to aggravation of symptoms.

The majority of cases were treated with high dose systemic corticosteroids, either enterally or parentally administered, prior to development or worsening of symptoms. Two cases (8%)

<u> </u>
7
×
~
-
ົທົ
ő
·=
0
2
۳.
ū
ö
ŏ
÷.
τ
0
Ö
ć
⇒
÷
5
-
2
Ð
2
5
,
õ
5
-
စ္ခာ
2
.2
ş
0
=
0
÷
ŝ
1
0
Õ
~
.≅
Ō
Ð
č
5
σ
÷
5
a
2
1
=
ي
τ
ā
<u> </u>
<u>e</u>
2
ž
5
~
2
Ħ
≥
5
ŝ
ē
ŝ
ğ
C
÷
0
ŝ
ă
Ĕ
E
0
<u>o</u>
Ħ
2
0
σ
ē
3
×.
۳.
5
ᅻ
a
œ.
-
8
22
.≃
2
-
0
-
÷
-
e
~
at
Tak

Table	1. Clinical features ar	nd outcomes of cases	with seve	re or fulminant ame	ebic colitis fo	ollowing treatment with	n corticosteroids,	, 1991–2016.				
Age (y)	Primary steroid indication	Co-morbid condition	Steroid form	Other immune modulating Rx	Country reported	Risk factor acquisition of amebiasis	Symptoms of amebic colitis	Area of gut involved	MTZ Rx	Surgery	Survival	Ref
60	Misdiagnosed UC	•	ш	Mesa	Japan	•	A, WL, D, H	ж	≻	z	≻	15
57	GVHD	BMT for AML	РО	FK508, ATG, MTX	Japan		AP, D, H, T	R, DC, TC	≻	z	۲	[16]
60	Multiple myeloma	CMV colitis	РО	•	Japan	Travel to endemic country*	D, H, F	SN	≻	z	z	[17]
48	Misdiagnosed UC		РО	Mesa	Japan		٩	РС	≻	۲	۲	[18]
55	Misdiagnosed UC	•	≥	5-ASA	Japan		D, H, AP. TM	NS	≻	≻	z	[19]
50	Nephrotic syndrome	Misdiagnosed UC; HIV; CMV/ fungal colitis	PO, IV	5-ASA, Mesa	Taiwan		H, P	РС	≻	≻	z	20
34	Interstitial pneumonia	AIDS; CMV colitis; MAC pneumonitis	NS	•	Taiwan		AP, D, P	U	≻	≻	≻	21
68	Misdiagnosed IBD	Alcoholism	РО	5-ASA	India	Endemic	D, H, AP, P	ЪС	≻	≻	z	22
31	Misdiagnosed lupus of the gut	Lupus; Salmonellosis	PO, IV	•	Singapore	Endemic	D, H, AP, F	РС	≻	z	z	[23]
35	Misdiagnosed abdominal TB	Hepatitis C infection	РО		Pakistan	Endemic	AP, D, F	IC, Am	≻	≻	Y	24
47	Misdiagnosed intestinal vasculitis	Lupus	PO, IV	Hydroxy-chloroquine	South Korea		AP, H, P	РС	≻	≻	≻	25
28	Misdiagnosed UC		РО		Spain	Travel to Philippines	D, H, F, WL	ЪС	≻	z	≻	26
33	Thrombotic thrombocytopenic purpura	Transfusion related acute lung injury	Sys	Plasmapheresis	Spain	Partner travelled to the Far East and North Africa	AP, D, H, P	U	≻	≻	≻	27
41	Misdiagnosed IBD		E, IV	5-ASA	Sweden	Travel to Brazil, Saudi Arabia, Sri Lanka, Tanzania	D, AP	R, DC, TC	≻	z	≻	28
99	Misdiagnosed IBD	HTN; psoriatic arthritis	PO, IV	MTX	Denmark	Travel to Bali	AP, D, H, WL	AC, TC, Am	≻	z	≻	[29]
39	Misdiagnosed CD		РО		Portugal	Travel to Angola	H, A, WL, AP, F	AC, C, LA	≻	z	۲	30
36	GVHD	BMT for AML; CONS sepsis	РО	MXT, CSa	Italy	Travel to North Africa	AP, D, H	TC, C, PS	≻	z	≻	31
ŝ	Pneumonia		РО		NSA		H, F	AC, LA	≻	z	≻	32
56	Misdiagnosed UC		IV, PO	5-ASA, Mesa	NSA	Travel to Mexico	D, H, AP, WL, P	PC, LA	≻	≻	≻	33
42	Liver transplant	Hepatitis C cirrhosis	РО	FK508, AZA	NSA	Well water	AP, D, T	AC, TC	≻	z	۲	34
-	Bronchiolitis	Streptococcus pyogenes bacteremia*	≥		American Samoa	Polynesian	D, P	Jejunum	≻	≻	≻	35
4	GVHD	BMT for AML; GNS	РО	CSa	Chile	Endemic	D, AP, F	н	≻	z	۲	36
15	GVHD	BMT for AML; CMV and fungal colitis	PO, IV	CSa	Chile	Endemic	D, AP, F	РС	≻	z	z	36
37	CD		IV, PO	5-ASA, 6-MP	Israel		F, H, D	AC	≻	≻	۲	37
* Con	tact with sex workers; A	l, anorexia; AC, Ascend	ing colon;	<i>Am</i> , ameboma; <i>AM</i> I	-, acute myel	oid leukemia; AP, abdor	ninal pain; ATG, a	ntithymocyte	globulir	ı; AZA, az	athioprine	 0
BMT,	bone marrow transplan	t; <i>CD</i> , Crohn disease; C	MV, cyton	negalovirus; CONS,	coagulase-ne	gative Staphylococcus	aureus; CSa, cycl	osporine; D, c	liarrhea 	t; <i>DC</i> , des	cending	
colon;	E, enema; F, tever; FK	508, tacrolimus; GNS; C	iram nega	tive sepsis; GVHD, g	Iratt-versus-h	ost disease; H, bloody s	stools; / <i>BD</i> , inflamr	natory bowel	disease	e; /C, ileoc	olic; //,	
mouth	erious; LA, irver abscess ; PS; proctosigmoid; R,	s; <i>INE</i> , meningoencepre rectum; <i>Rx</i> , treatment;	T, tenesmi	a, mesalarnine; <i>mi X</i> , us: <i>TB</i> . tuberculosis:	TC. transver	e; ivi i z, metronidazole; se colon: <i>TM</i> , toxic med	w, no; wo, not stat acolon: <i>U</i> C. ulcera	teu; ~, periora ative colitis: <i>W</i>	L. weid	o, paricoli ht loss; Y	us; r-0, p ves; 5-A	y SA,

sulfasalazine; 6-MP, mercaptopurine doi:10.1371/journal.pntd.0004879.t001 however, seemed to worsen after treatment with corticosteroid enemas [15,28]. Corticosteroid preparation, dose administered and exact duration of time prior to worsening were inconsistently reported, prohibiting calculation of average cumulative dosages.

## **Clinical features**

All cases met the study definition of either severe or fulminant amebic colitis and presented with a combination of acute worsening of abdominal pain, diarrhea, and /or bloody stools/ dysentery. Eight (33%) cases were complicated by intestinal perforation [18,20–22,25,27,33,35]. Perforation occurred in both of the cases with HIV, but none of the transplant cases. The odds of perforation did not differ significantly by prior misdiagnosis, CMV co-infection or treatment with additional immune modulating therapies. Associated amebic liver abscesses were found in three (12.5%) cases [30,32,33]. Central nervous system dissemination occurred in one case (4%), a one-year-old child with jejenal perforation, liver abscess and seizures secondary to meningoencephalitis (trophozoites demonstrated by microscopy in the cerebrospinal fluid) [35]. Amebomas were found in two cases (8%) [24,29]. One case (4%) developed toxic megacolon [19]. One case (4%) believed to have both Crohn disease and amebic colitis developed a rectovaginal fistula [37].

Most cases (13 of 24, 54%) had involvement of multiple areas of bowel, including pancolitis in seven cases (29%) [18,20,22,23,25,26,33]. Proctosigmoid involvement was documented in four cases (17%) [15,16,28,31], descending colon in two cases (8%)[16,28], transverse colon in five cases (21%)[16,28,29,31,34], ascending colon in five cases (21%) [29–32,34], cecal/ ileo-cecal in five cases (21%) [21,24,27,30,31], jejenal in one case (4%) [35] and exact location of colitis not specified in two cases (8%) [17,19].

#### Laboratory diagnosis

A diagnosis of amebiasis was established in 22 of 24 (92%) cases by microscopy, serology and/ or histology. Post-mortem examination established the diagnosis in one case, revealing E. histo*lytica* in the ulcerated colonic wall [19]. Presumptive diagnosis was made in another case, on the clinical basis of worsening symptoms with anti-mycobacterial therapy and corticosteroids followed by improvement with metronidazole, and supported by suggestive ulcerations on histology. Leukocytosis was reported in 10 of 13 cases (77%). Stool microscopy identified Entamoeba cysts and/or trophozoites in 8 of 16 (50%) cases. Methodology used and number of stools submitted was variable, and not consistently reported. Four of the 8 cases with positive microscopy (50%) had at least one other positive study. Serology was positive in 10 of 15 (67%) cases. Of those with positive serology, 7 of 10 cases (70%) had at least one other positive study. The 5 cases with negative serology included both cases who underwent stem cell transplantation [16,31], a case with AIDS [21], a case with Lupus [23] and a case with Crohn disease [37]. The type of serologic assay performed was not consistently reported. Submitted histology from biopsy at colonoscopy showed tissue trophozoites in 7 of 15 (47%) of cases. In cases who underwent surgery, submitted surgical pathology yielded direct evidence of amebiasis in 8 of 9 (89%) of cases. The use of a direct antigen test was reported in a single case [26].

#### Outcomes

All cases were reported to have received appropriate therapy with metronidazole. Most cases also reported use of a luminal agent, such as paromomycin, though not always commercially available [16]. One case did not respond to treatment until corticosteroids were withdrawn [36]. Eleven (46%) underwent surgical treatment of their disease (Table 1). There were 6 fatalities (25%), half underwent surgical intervention prior to death (Table 1). The odds of death

did not differ significantly by prior misdiagnosis, transplant status, HIV co-infection, CMV co-infection, perforation, surgery or presence of additional immune modulating therapies.

#### Discussion

We identified 24 cases of severe and fulminant amebic colitis treated with corticosteroids over the past 25 years. To our knowledge, this is the largest comprehensive report of such cases. It is interesting that nearly one-third of cases were in returning travelers. Travel to South and South East Asia, Africa, the Middle East and South America were listed as risk factors for European travelers. Travel to Mexico was significant for one US traveler. Data analyzed from the GeoSentinel Surveillance Network, a worldwide network that performs surveillance and monitoring of travel related illnesses, showed that E. histolytica is the third most frequently isolated pathogen among returning travelers with infectious gastrointestinal disease, accounting for 12.5% of all microbiologically confirmed cases, with an estimated rate of 14.0 per 1000 returned travelers [38]. Rates varied by region of travel and category of traveler, with the highest rates of amebiasis reported in travelers to South Asia, the Middle East and South America, similar to our findings, and among "missionary/ volunteering" travelers [38]. Amebiasis was not considered in any of our reported travelers when they initially presented leading to misdiagnosis. The low incidence of amebiasis in industrialized countries can lead to unfamiliarity with the clinical presentation. Given the significant morbidity and mortality associated with fulminant disease, it is important for providers to enquire about travel history and screen travelers to and migrants from endemic regions for amebiasis, using an appropriate test, such as the fecal antigen test, even if travel occurred in the distant past [13]. Providers should also inquire about the travel history of close household and sexual contacts [4,39]. The utility of this is demonstrated in the case who acquired amebiasis after her partner travelled to the Far East and North Africa [40].

While the incidence of fulminant amebic colitis is likely to be more common in endemic countries given the high frequency of asymptomatic intestinal carriage in many indigenous populations, it is probable that these episodes are not being published in the literature, leading to underrepresentation in our compiled series. Even among cases included from countries with higher endemicity, amebiasis was often not initially suspected. Inflammatory bowel disease was the most frequently misdiagnosed condition at presentation. The diagnostic dilemma here is that many of the symptoms of amebic colitis overlap with symptoms of inflammatory bowel disease may also have amebiasis, a considerable diagnostic challenge in endemic countries [41-44]. In countries where amebiasis is endemic, there should be a low threshold for suspecting in patients who present with symptoms of inflammatory bowel disease. All patients with a new diagnosis of inflammatory bowel disease should be screened for amebiasis with a stool study for fecal antigen testing or serum for amebic serology, especially if they reside in or have a history of travel to an endemic area.

It is important to note that while in most patients the cause of colitis was initially misdiagnosed; almost 40% of cases did not have any gastrointestinal symptoms prior to initiation of corticosteroids, proving the need to consider asymptomatic intestinal carriage prior to corticosteroid administration as well. The reason that only a subset of people infected with amebiasis develop clinical disease is poorly understood and challenging to study, given the complexity of host-amebic interactions. Ultimately, it is the combination of amebic virulence factors and destructive host inflammatory responses that mediate tissue injury seen with amebic colitis.<sup>47</sup> Cell-mediated responses appear to provide some protection against amebiasis [45]. For example, macrophages and neutrophils activated by cell mediated interferon- $\gamma$  kill trophozoites in vitro [46,47]. In addition, children with higher levels of interferon- $\gamma$  production have lower susceptibility to subsequent symptomatic amebiasis [48]. The development of fulminant amebic colitis in patients treated with corticosteroids further supports the importance of cell-mediated immunity in controlling *E. histolytica* infection. Rodent models of amebic colitis have demonstrated causal relationships between corticosteroids and exacerbation of amebic disease [49–51]. However the exact mechanism leading to such rapid and expansive intestinal inflammation and necrosis is unknown. That said, *E. histolytica* encodes for a macrophage migration inhibitory factor (MIF) homolog which has been shown to inhibit the anti-inflammatory responses of corticosteroids in vitro, potentially promoting a pro-inflammatory state [52].

Patients with amebic colitis may present initially with acute onset abdominal pain, diarrhea and bloody stools or less commonly with a more chronic course of diarrhea, weight loss and abdominal pain. Many of these symptoms overlap with those of inflammatory bowel disease, and even stool inflammatory markers, imaging, endoscopic findings and lesion distribution in amebic colitis can be non-distinguishing and difficult to differentiate from inflammatory bowel disease [53]. A number of diagnostic modalities, including stool studies, histology, and serology are available to assist, though a combination of techniques is often used to establish the presence of infection.

Cysts and trophozoites (with or without hemophagocytosis) can be visualized by an experienced eye using stool microscopy, but this test lacks specificity. Advances in molecular epidemiology have revealed that three other Entamoeba species associated with human infection are morphologically indistinguishable from E. histolytica (including E. dispar, E. moshokovskii and *E. bangladeshi*) and cannot be differentiated from *E. histolytica* by microscopy [54]. In addition, stool excretion is variable and multiple stool samples must be submitted to maximize the chance of visualization by microscopy. At best, the sensitivity of stool microscopy is only 60%, consistent with the sensitivity of 50% noted in our study, yet this modality remains the most widely used test of diagnosis worldwide [55]. More sensitive methods are available and stool microscopy is no longer recommended for diagnosis. Several antigen detection tests specific for E. histolytica have been recently developed, such as the Techlab E. HISTOLYTICA II test (detects E. histolytica-derived Gal/ GalNAc-specific lectin), offering high sensitivity and specificity, up to 94% and 100% respectively [56-58]. Antigen detection tests are simple to perform, rapid, can be tested in batches, and may even be combined to simultaneously detect multiple parasitic enteropathogens, but unfortunately, despite these advantages, remain underutilized, as evidenced by only a single report in our study documenting use of a rapid antigen detection test [26,59]. Other tests, such as rapid detection of lectin antigen in stool and serum are under development [57]. Polymerase chain reaction (PCR) methods also show high sensitivity and specificity, but are mostly available as research tools, though may prove to be more useful as commercial tests in the future [60]. Detection of antibodies by serologic assay (indirect fluorescent, counter immunoelectrophoresis or enzyme linked immunosorbent assay) is often possible by the time of presentation. Serology may be a useful adjunct to stool studies. The sensitivity of serology ranges from 60-90%, consistent with the sensitivity of 67% found in our study. The utility of serology in endemic areas with high seroprevalence is limited, however, as serology may remain positive years after infection. In more severe cases, histology obtained by biopsy at colonoscopy (Fig 2), flexible sigmoidoscopy or surgical resection may show ulcers (e.g., classic flask-shaped ulceration), sometimes with cysts and trophozoites found at the lesion edge, and proved useful in 47% of the cases undergoing endoscopy and 89% of patients undergoing surgical resection in our cases studied.

All patients with evidence of *E. histolytica* infection should receive treatment, even if asymptomatic, to prevent progression, especially as the potential always exists for unknown future need to administer corticosteroids. For amebic colitis and other symptomatic forms of disease,



**Fig 2.** Invasion of colonic mucosa by amebic trophozoites with resultant inflammatory response. Colonic biopsy taken from a 49-year-old US male migrant from Mexico, diagnosed with severe amebic colitis after receiving high dose dexamethasone during management of subarachnoid hemorrhage (Courtesy of William A. Petri, Jr., University of Virginia)

doi:10.1371/journal.pntd.0004879.g002

a two-drug approach is taken with an amebicidal agent, such as the nitroimidazole metronidazole [61], to eliminate invading trophozoites and a luminal cysticidal agent such as the nonabsorbable aminoglycoside paromomycin to eradicate intraluminal carriage of cysts [62]. If screening reveals asymptomatic carriage, then treatment with a luminal agent is sufficient to prevent progression of disease and transmission of infection [5]. Patients who are severely ill with amebic colitis should be hospitalized; supportive and intensive care should be provided as indicated. The addition of broad spectrum antibiotics may be required in the presence of peritonitis. Urgent surgical intervention is needed in the setting of bowel perforation. Toxic megacolon or extensive necrosis may require colectomy. In our study, almost half of cases required surgical intervention and a quarter died, despite appropriate antimicrobial therapy, fortifying the high morbidity and mortality associated with fulminant amebic colitis and the urgent need for improved therapeutic options. We did not find prior transplant, HIV infection, CMV infection, perforation, surgery or use of additional immune modulating therapies to be associated with the odds of death.

While there is no vaccine available to prevent amebiasis, the demonstration of at least partially protective host humoral and cell-mediated immune responses supports the need to continue efforts towards vaccine development [63]. In the meantime, travelers to endemic areas should be given advice on the avoidance of risk factors for acquisition of infection, and counselled on the use of proper hand hygiene, food and water precautions and avoidance of fecal exposure during sexual activity [64].

#### Table 2. Summary of findings and recommendations.

Patients with either symptomatic or asymptomatic intestinal amebiasis treated with corticosteroid therapy are at high risk of developing the potentially fatal complication of fulminant amebic colitis.

Infection with *E. histolytica* should be considered prior to the diagnosis of inflammatory bowel disease, and for subsequent exacerbations.

Travel history of patients, their close house hold and sexual contacts should be obtained prior to initiation of systemic corticosteroids. Patients residing in or with travel history to endemic parts of the world, such as South and Southeast Asia, Africa, Central America, South America and Mexico, should be screened for amebiasis with tests that afford the highest level of sensitivity and specificity available.

All patients with evidence of *E. histolytica* infection should be treated appropriately prior to initiating corticosteroids to prevent fulminant amebic colitis. Amebic colitis should be treated with metronidazole followed by a luminal agent, such as paromomycin. Treatment with a luminal agent alone is sufficient for patients with asymptomatic intestinal amebiasis.

Research efforts are needed to develop both interventions to prevent amebic colitis, and additional therapies to treat fulminant amebic colitis are needed to improve outcomes.

doi:10.1371/journal.pntd.0004879.t002

There are several limitations to our study. Severe and fulminant forms of colitis in patients treated with steroids are likely vastly underreported in the literature. Due to the retrospective nature and extraction of data from prior reports, incomplete information was collected from case descriptions. The small sample size may have limited the power to detect significant associations.

# Conclusions

Our study describes 24 cases of patients who developed severe and fulminant colitis following treatment with corticosteroid therapy, emphasizing the high morbidity and mortality associated with this condition and identifying knowledge gaps that must be addressed in the future. Efforts should be made to improve familiarity of this diagnosis among health care providers. In endemic countries, infection with *E. histolytica* must be excluded prior to the diagnosis of inflammatory bowel disease and treatment with corticosteroids administered systemically or by enema. Empiric treatment of amebiasis should be considered in situations where diagnostic testing is not readily available. In non-endemic countries, prior to starting corticosteroids or other immune suppressive therapy, travel history of the patient, close household and sexual contacts should be obtained. For those who have traveled to or migrated from endemic countries, screening for amebiasis should be carried out and those with positive test should be treated (Table 2). The most sensitive and specific test(s) available should be chosen to satisfactorily exclude. Consideration of *E. histolytica* in such situations will help to avoid a delay in diagnosis and potential death [65].

# **Supporting Information**

**S1 Checklist. STROBE Checklist.** (DOCX)

# Acknowledgments

We thank William A. Petri, Jr. for helpful advice and support.

# **Author Contributions**

Conceived and designed the experiments: SM. Analyzed the data: SM DAS. Wrote the paper: SM DAS.

# References

- Lozano R, Naghavi M, Foreman K, Lim S, Shibuya K, Aboyans V, et al. Global and regional mortality from 235 causes of death for 20 age groups in 1990 and 2010: a systematic analysis for the Global Burden of Disease Study 2010. Lancet. 2012; 380: 2095–2128. doi: <u>10.1016/S0140-6736(12)61728-0</u> PMID: <u>23245604</u>
- 2. Stanley SL Jr. Amoebiasis. Lancet. 2003; 361: 1025–1034. PMID: 12660071
- Kotloff KL, Nataro JP, Blackwelder WC, Nasrin D, Farag TH, Panchalingam S, et al. Burden and aetiology of diarrhoeal disease in infants and young children in developing countries (the Global Enteric Multicenter Study, GEMS): a prospective, case-control study. Lancet. 2013; 382: 209–222. doi: <u>10.1016/S0140-6736(13)60844-2</u> PMID: <u>23680352</u>
- Salit IE, Khairnar K, Gough K, Pillai DR. A possible cluster of sexually transmitted Entamoeba histolytica: genetic analysis of a highly virulent strain. Clin Infect Dis. 2009; 49: 346–353. doi: <u>10.1086/600298</u> PMID: <u>19580413</u>
- 5. Haque R, Huston CD, Hughes M, Houpt E, Petri WA. Amebiasis. N Engl J Med. 2003; 348: 1565–1573. PMID: <u>12700377</u>

- 6. Alvi AR, Jawad A, Fazal F, Sayyed R. Fulminant amoebic colitis: a rare fierce presentation of a common pathology. Trop Doct. 2013; 43: 80–82. doi: 10.1177/0049475513491725 PMID: 23796678
- Athie-Gutierrez C, Rodea-Rosas H, Guizar-Bermudez C, Alcantara A, Montalvo-Jave EE. Evolution of surgical treatment of amebiasis-associated colon perforation. J Gastrointest Surg. 2010; 14: 82–87. doi: <u>10.1007/s11605-009-1036-y</u> PMID: <u>19774426</u>
- Nisheena R, Ananthamurthy A, Inchara YK. Fulminant amebic colitis: a study of six cases. Indian J Pathol Microbiol. 2009; 52: 370–373. doi: <u>10.4103/0377-4929.54997</u> PMID: <u>19679964</u>
- Chaturvedi R, Gupte PA, Joshi AS. Fulminant amoebic colitis: a clinicopathological study of 30 cases. Postgrad Med J. 2015; 91: 200–205. doi: 10.1136/postgradmedj-2014-132597 PMID: 25748520
- Takahashi T, Gamboa-Dominguez A, Gomez-Mendez TJ, Remes JM, Rembis V, Martinez-Gonzalez D, et al. Fulminant amebic colitis: analysis of 55 cases. Dis Colon Rectum. 1997; 40: 1362–1367. PMID: 9369114
- Kanani SR, Knight R. Amoebic dysentery precipitated by corticosteroids. Br Med J. 1969; 3: 114. PMID: <u>4307124</u>
- 12. Goodwin CS. Amoebic colitis exacerbated by steroids. Br Med J. 1969; 3: 174.
- 13. Stuiver PC, Goud TJ. Corticosteroids and liver amoebiasis. Br Med J. 1978; 2: 394–395. PMID: 687936
- Petri WA Jr, Singh U. Diagnosis and management of amebiasis. Clin Infect Dis. 1999; 29: 1117–1125. PMID: 10524950
- Den Y, Kinoshita J, Deshpande GA, Hiraoka E. Amoebiasis masquerading as inflammatory bowel disease. BMJ Case Rep. 2015; 2015: doi: 10.1136/bcr-2015-212102
- Numata A, Itabashi M, Kishimoto K, Motohashi K, Hagihara M, Kuwabara H, et al. Intestinal amoebiasis in a patient with acute graft-versus-host disease after allogeneic bone marrow transplantation successfully treated by metronidazole. Transpl Infect Dis. 2015; 17: 886–889. doi: <u>10.1111/tid.12460</u> PMID: <u>26426525</u>
- Kobayashi CI, Yamamoto G, Hayashi A, Ota S, Imai Y, Fukayama M, et al. Fatal amebic colitis after high-dose dexamethasone therapy for newly diagnosed multiple myeloma. Ann Hematol. 2011; 90: 225–226. doi: 10.1007/s00277-010-0984-3 PMID: 20467744
- Ishida H, Inokuma S, Murata N, Hashimoto D, Satoh K, Ohta S. Fulminant amoebic colitis with perforation successfully treated by staged surgery: a case report. J Gastroenterol. 2003; 38: 92–96. PMID: 12560929
- Yamamoto J, Matsumoto K, Suzuki H. Amebic colitis in inflammatory bowel disease. J Clin Gastroenterol. 1992; 14: 91–92. PMID: <u>1556420</u>
- 20. Weng,Meng-Tzu, Wei,Shu-Chen, Leong,Yew-Loong, Tung,Chien-Chih, Shun,Chia-Tung, Wong,Jau-Min.
- Tsai HC, Lee SS, Wann SR, Chen YS, Chen ER, Yen CM, et al. Colon perforation with peritonitis in an acquired immunodeficiency syndrome patient due to cytomegalovirus and amoebic colitis. J Formos Med Assoc. 2005; 104: 839–842. PMID: 16496064
- Gupta SS, Singh O, Shukla S, Raj MK. Acute fulminant necrotizing amoebic colitis: a rare and fatal complication of amoebiasis: a case report. Cases J. 2009; 2: 6557-1626-2-6557.
- 23. Tai ES, Fong KY. Fatal amoebic colitis in a patient with SLE: a case report and review of the literature. Lupus. 1997; 6: 610–612. PMID: <u>9302665</u>
- 24. Humayun M, Badshah A, Khan I. CHRONIC CAECAL AMEBIASIS MIMICKING FEATURES SUG-GESTIVE OF ABDOMINAL TUBERCULOSIS AND CROHN'S DISEASE: A CASE REPORT. 2013; 5: 103.
- Lee J, Jung HS, Nam HC, Kwok SK, Ju JH, Park KS, et al. Fulminant amoebic colitis mimicking intestinal vasculitis in a patient with systemic lupus erythematosus. Lupus. 2012; 21: 1351–1355. PMID: 22570337
- Prim N, Escamilla P, Sole R, Llovet T, Soriano G, Munoz C. Risk of underdiagnosing amebic dysentery due to false-negative Entamoeba histolytica antigen detection. Diagn Microbiol Infect Dis. 2012; 73: 372–373. doi: <u>10.1016/j.diagmicrobio.2012.04.012</u> PMID: <u>22683092</u>
- Forteza A, Ballester Ruiz C, Visvesvara G, Muncunill Ribas J, Besalduch Vidal J, Couce ME. Resolution of refractory thrombotic thrombocytopenic purpura (TTP) after successful treatment of a fulminant colitis due to Entamoeba histolytica. Gastroenterol Hepatol. 2013; 36: 294–295. doi: <u>10.1016/j.gastrohep.2012.11.004</u> PMID: <u>23453561</u>
- Larsson PA, Olling S, Darle N. Amebic colitis presenting as acute inflammatory bowel disease. Case report. Eur J Surg. 1991; 157: 553–555. PMID: <u>1683582</u>

- Mogensen TH, Christiansen JJ, Eivindson MV, Larsen CS, Tottrup A. Misdiagnosed amoebic colitis leading to severe dysentery and necrotizing colitis—report of a case and review of the literature. Scand J Infect Dis. 2014; 46: 235–239. doi: 10.3109/00365548.2013.871646 PMID: 24450838
- Caetano AC, Goncalves B, Rolanda C. Rare complication: what kind of colitis? Gastroenterology. 2013; 145: e7–8.
- **31.** Bavaro P, Di Girolamo G, Di Bartolomeo P, Angrilli F, Olioso P, Papalinetti G, et al. Amebiasis after bone marrow transplantation. Bone Marrow Transplant. 1994; 13: 213–214. PMID: <u>8205092</u>
- Ha K.The case of mistaken identity: Entamoeba histolytica presenting as community acquired pneumonia. American Journal of Gastroenterology. 2016; 110:S386.
- Abbas MA, Mulligan DC, Ramzan NN, Blair JE, Smilack JD, Shapiro MS, et al. Colonic perforation in unsuspected amebic colitis. Dig Dis Sci. 2000; 45: 1836–1841. PMID: <u>11052328</u>
- Palau LA, Kemmerly SA. First report of invasive amebiasis in an organ transplant recipient. Transplantation. 1997; 64: 936–937. PMID: <u>9326427</u>
- Goh LM, Marrone JR. Entamoeba histolytica meningoencephalitis diagnosed by trophozoites in cerebrospinal fluid. New Microbes New Infect. 2013; 1: 16–17. doi: 10.1002/2052-2975.6 PMID: 25356319
- Perret C, Harris PR, Rivera M, Vial P, Duarte I, Barriga F. Refractory enteric amebiasis in pediatric patients with acute graft-versus-host disease after allogeneic bone marrow transplantation. J Pediatr Gastroenterol Nutr. 2000; 31: 86–90. PMID: 10896079
- Lysy J, Zimmerman J, Sherman Y, Feigin R, Ligumsky M. Crohn's colitis complicated by superimposed invasive amebic colitis. Am J Gastroenterol. 1991; 86: 1063–1065. PMID: <u>1858742</u>
- Swaminathan A, Torresi J, Schlagenhauf P, Thursky K, Wilder-Smith A, Connor BA, et al. A global study of pathogens and host risk factors associated with infectious gastrointestinal disease in returned international travellers. J Infect. 2009; 59: 19–27. doi: <u>10.1016/j.jinf.2009.05.008</u> PMID: <u>19552961</u>
- Vreden SG, Visser LG, Verweij JJ, Blotkamp J, Stuiver PC, Aguirre A, et al. Outbreak of amebiasis in a family in The Netherlands. Clin Infect Dis. 2000; 31: 1101–1104. PMID: <u>11049797</u>
- 40. Forteza A, Ballester Ruiz C, Visvesvara G, Muncunill Ribas J, Besalduch Vidal J, Couce ME. Resolution of refractory thrombotic thrombocytopenic purpura (TTP) after successful treatment of a fulminant colitis due to Entamoeba histolytica. Gastroenterol Hepatol. 2013; 36: 294–295. doi: <u>10.1016/j.gastrohep.2012.11.004</u> PMID: <u>23453561</u>
- Ozin Y, Kilic MZ, Nadir I, Tayfur O, Ertas A, Ulker A, et al. Presence and diagnosis of amebic infestation in Turkish patients with active ulcerative colitis. Eur J Intern Med. 2009; 20: 545–547. doi: <u>10.1016/j.</u> ejim.2009.05.014 PMID: <u>19712863</u>
- Vukobrat-Bijedic Z, Husic-Selimovic A, Bijedic N, Bjelogrlic I, Djuran A. Intestinal amebiasis in a group of patients with ulcerative colitis: influence on clinical course of the disease. Med Arch. 2013; 67: 10– 12. PMID: 23678829
- Tozun N, Atug O, Imeryuz N, Hamzaoglu HO, Tiftikci A, Parlak E, et al. Clinical characteristics of inflammatory bowel disease in Turkey: a multicenter epidemiologic survey. J Clin Gastroenterol. 2009; 43: 51–57. doi: 10.1097/MCG.0b013e3181574636 PMID: 18724251
- Yamamoto-Furusho JK, Torijano-Carrera E. Intestinal protozoa infections among patients with ulcerative colitis: prevalence and impact on clinical disease course. Digestion. 2010; 82: 18–23. doi: <u>10.1159/</u> 000273871 PMID: <u>20145404</u>
- 45. Moonah SN, Jiang NM, Petri WA Jr. Host immune response to intestinal amebiasis. PLoS Pathog. 2013; 9: e1003489. doi: 10.1371/journal.ppat.1003489 PMID: 23990778
- 46. Denis M, Chadee K. Human neutrophils activated by interferon-gamma and tumour necrosis factoralpha kill Entamoeba histolytica trophozoites in vitro. J Leukoc Biol. 1989; 46: 270–274. PMID: 2547889
- Salata RA, Pearson RD, Ravdin JI. Interaction of human leukocytes and Entamoeba histolytica. Killing of virulent amebae by the activated macrophage. J Clin Invest. 1985; 76: 491–499. PMID: <u>2863284</u>
- Haque R, Mondal D, Shu J, Roy S, Kabir M, Davis AN, et al. Correlation of interferon-gamma production by peripheral blood mononuclear cells with childhood malnutrition and susceptibility to amebiasis. Am J Trop Med Hyg. 2007; 76: 340–344. PMID: <u>17297046</u>
- Asgharpour A, Gilchrist C, Baba D, Hamano S, Houpt E. Resistance to intestinal Entamoeba histolytica infection is conferred by innate immunity and Gr-1+ cells. Infect Immun. 2005; 73: 4522–4529. PMID: 16040963
- TEODOROVIC S, INGALLS JW, GREENBERG L. Effects of corticosteroids on experimental amoebiasis. Nature. 1963; 197: 86–87. PMID: <u>13980603</u>
- Vinayak VK, Chitkara NL, Chhuttani PN. Effect of corticosteroid and irradiation on caecal amoebic infection in rats. Trans R Soc Trop Med Hyg. 1979; 73: 266–268. PMID: <u>473322</u>

- Moonah SN, Abhyankar MM, Haque R, Petri WA Jr. The macrophage migration inhibitory factor homolog of Entamoeba histolytica binds to and immunomodulates host macrophages. Infect Immun. 2014; 82: 3523–3530. doi: 10.1128/IAI.01812-14 PMID: 24818664
- Horiki N, Furukawa K, Kitade T, Sakuno T, Katsurahara M, Harada T, et al. Endoscopic findings and lesion distribution in amebic colitis. J Infect Chemother. 2015; 21: 444–448. doi: <u>10.1016/j.jiac.2015.02.</u> 004 PMID: 25787830
- 54. Ali IK. Intestinal amebae. Clin Lab Med. 2015; 35: 393–422. doi: <u>10.1016/j.cll.2015.02.009</u> PMID: <u>26004649</u>
- Singh A, Houpt E, Petri WA. Rapid Diagnosis of Intestinal Parasitic Protozoa, with a Focus on Entamoeba histolytica. Interdiscip Perspect Infect Dis. 2009; 2009: 547090. doi: <u>10.1155/2009/547090</u> PMID: <u>19584941</u>
- 56. Haque R, Mollah NU, Ali IK, Alam K, Eubanks A, Lyerly D, et al. Diagnosis of amebic liver abscess and intestinal infection with the TechLab Entamoeba histolytica II antigen detection and antibody tests. J Clin Microbiol. 2000; 38: 3235–3239. PMID: <u>10970364</u>
- 57. Haque R, Petri WA Jr. Diagnosis of amebiasis in Bangladesh. Arch Med Res. 2006; 37: 273–276. PMID: <u>16380331</u>
- Nunez YO, Fernandez MA, Torres-Nunez D, Silva JA, Montano I, Maestre JL, et al. Multiplex polymerase chain reaction amplification and differentiation of Entamoeba histolytica and Entamoeba dispar DNA from stool samples. Am J Trop Med Hyg. 2001; 64: 293–297. PMID: <u>11463120</u>
- 59. Christy NC, Hencke JD, Escueta-De Cadiz A, Nazib F, von Thien H, Yagita K, et al. Multisite performance evaluation of an enzyme-linked immunosorbent assay for detection of Giardia, Cryptosporidium, and Entamoeba histolytica antigens in human stool. J Clin Microbiol. 2012; 50: 1762–1763. doi: <u>10.1128/JCM.06483-11</u> PMID: <u>22378909</u>
- Verweij JJ, van Lieshout L. Intestinal parasitic infections in an industrialized country; a new focus on children with better DNA-based diagnostics. Parasitology. 2011; 138: 1492–1498. doi: <u>10.1017/</u> <u>S0031182011001211</u> PMID: <u>21859503</u>
- Powell SJ, Wilmot AJ, Elsdon-Dew R. Further trials of metronidazole in amoebic dysentery and amoebic liver abscess. Ann Trop Med Parasitol. 1967; 61: 511–514. PMID: <u>4336147</u>
- Pamba HO, Estambale BB, Chunge CN, Donno L. Comparative study of aminosidine, etophamide and nimorazole, alone or in combination, in the treatment of intestinal amoebiasis in Kenya. Eur J Clin Pharmacol. 1990; 39: 353–357. PMID: 2076717
- **63.** Quach J, St-Pierre J, Chadee K. The future for vaccine development against Entamoeba histolytica. Hum Vaccin Immunother. 2014; 10: 1514–1521. doi: <u>10.4161/hv.27796</u> PMID: <u>24504133</u>
- 64. [Anonymous]. Centers for Disease Control and Prevention. CDC Health Information for International Travel 2016. New York: Oxford University Press; 2016.
- Poetker DM, Smith TL. What rhinologists and allergists should know about the medico-legal implications of corticosteroid use: a review of the literature. Int Forum Allergy Rhinol. 2012; 2: 95–103. doi: <u>10.</u> <u>1002/alr.21016</u> PMID: <u>22253235</u>